Author’s response to reviews

Title: Oncological and functional results after surgical treatment of bone metastases at the proximal femur

Authors:
Grzegorz Guzik (grzegorz.guzik@vp.pl)

Version: 1 Date: 09 Aug 2017

Author’s response to reviews:

Dear Editors

I would like to thanks Editors and Reviewers for valuable comments which improved the quality of my article. All comments were considered in the revised manuscript.

BSUR-D-17-00303

Oncological and functional results after surgical treatment of bone metastases at the proximal femur

Grzegorz Guzik

BMC Surgery

Editor's comments:

Your manuscript will be send out for peer-review when the following issues are properly solved.

1. The Abstract should not exceed 350 words.
   • Background: the context and purpose of the study
   • Methods: how the study was performed and statistical tests used
   • Results: the main findings
   • Conclusions: brief summary and potential implications
2. The general information included in the Methods section should be moved to the Results section.

List of changes:

1. I changed the Abstract as Editor suggest – see in text

2. I changed the text in Methods and Results section as Editor suggest – see in text.

Once more thanks for Yours help and comments

Yours faithfully

Grzegorz Guzik